Author's response to reviews

Title: Challenges in identifying barriers to adoption using questionnaires in a theory-based implementation study: case study on postnatal depression recommendations

Authors:

Andria Hanbury (andria.hanbury@york.ac.uk)
Farley Katherine (katherine.farley@york.ac.uk)
Carl Thompson (carl.thompson@york.ac.uk)
Duncan Chambers (duncan.chambers@york.ac.uk)
Paul M Wilson (paul.wilson@york.ac.uk)

Version: 2 Date: 12 March 2012

Author's response to reviews:

Dear Editor,

Please accept our apologies for the oversight in submitting our article for peer review without referencing the research ethics approval that we have for this study. This revised version of our article has had the relevant details added in at the start of the methods section.

Yours sincerely,

Andria Hanbury